Author's response to reviews

Title: Immune responses in patients with HIV infection after vaccination with recombinant Hepatitis B virus vaccine

Authors:

Neelam Pasricha (pgimer@chd.nic.in)
Usha Datta (pgimer@chd.nic.in)
Yogesh Chawla (pgimer@chd.nic.in)
Surjit Singh (pgimer@chd.nic.in)
Sunil K Arora (pgimer@chd.nic.in)
Archana Sud (pgimer@chd.nic.in)
Ranjana W Minz (pgimer@chd.nic.in)
Biman Saikia (bimansaikia@hotmail.com)
Haqeeqat Singh (pgimer@chd.nic.in)
Isaac James (pgimer@chd.nic.in)
Shobha Sehgal (pgimer@chd.nic.in)

Version: 3 Date: 13 December 2005

Author's response to reviews:

The following modifications have been made as suggested.

Title and Abstract:
Full stop removed from title both in the upload page and the manuscript.
Author affiliation list modified as suggested.
Title and Abstract are identical in the manuscript and the submission system.
Author list is written identically in order.
Keyword section has been removed.
Abstract has been restructured. Abstract inserted in the main manuscript.
Introduction:
Renamed as Background
Sentence case has been changed where suggested.
Conclusion section inserted after Discussion.
Competing interest section has been addressed.
Author contribution section added. (SS3 is Surjit Singh and SS1 is Shobha Sehgal).
References are cited using square brackets before full stop.
Author names added and term "et al" removed (Ref. 33 and 34).
Figure 2A is submitted as Fig. 2, Figure 2B as Fig. 3 and Figure 3 is changed to Fig 4. (The figure numbers in the main manuscript has been changed accordingly).
Figure titles and numbers have been removed from the figures and necessary changes made in the figure legend section.
Figures are cropped as closely as possible.
No additional files have been submitted.